Author's response to reviews

Title: Clinical Validation of Robot Simulation of Toothbrushing - Comparative Plaque Removal Efficacy

Authors:

Tomas Lang (info@ormed.net)
Sebastian Staufer (staufer@zahnarzt-ol.de)
Barbara Jennes (barbara.jennes@t-online.de)
Peter Gaengler (peter.gaengler@uni-wh.de)

Version: 4 Date: 19 June 2014

Author's response to reviews:

Dear editor,

we changed the heading in the abstract to Background, we included the conclusion section after the discussion, we removed the figures from the main manuscript but left the figure legends, and finally, removed the tables that had been uploaded as figures.

Best regards and kindest wishes

Peter Gängler and Tomas Lang